Author’s response to reviews

Title: Addressing gaps in surgical skills training by means of low-cost simulation at Muhimbili University in Tanzania

Authors:

Stephanie Taché (stache@globalhealth.ucsf.edu)
Naboth Mbembati (nmbembati@muhas.ac.tz)
Nell Marshall (nell_marshall@yahoo.com)
Frank Tendick (frank.tendick@gmail.com)
Charles Mkony (cmkony@muhas.ac.tz)
Patricia O’Sullivan (OsullivanP@medsch.ucsf.edu)

Version: 5 Date: 23 July 2009

Author’s response to reviews:

Dear Editor,

We have modified the major and minor reference changes that you outlined in your email correspondence.